Author's response to reviews

Title: Improving the adherence of type 2 diabetes mellitus patients with pharmacy care: a systematic review of randomized controlled trials

Authors:

Sunya-Lee Antoine (sunya-lee.antoine@uni-wh.de)
Dawid Pieper (dawid.pieper@uni-wh.de)
Tim Mathes (tim.mathes@uni-wh.de)
Michaela Eikermann (michaela.eikermann@uni-wh.de)

Version: 2
Date: 29 April 2014

Author's response to reviews:

Dear Editors,

the changes to the manuscripts have been made as requested. These include the addition of line numbering, the deletion of inser page breaks and the changed heading from "Contribution" to "Author`s Contribution".

Sincerely,
Sunya-Lee Antoine